Author's response to reviews

Title: A Cluster-Randomized, Placebo-controlled, Maternal Vitamin A or Beta-Carotene Supplementation Trial in Bangladesh: Design and Methods

Authors:

Alain B Labrique (alabriqu@jhsph.edu)
Parul Christian (pchristi@jhsph.edu)
Rolf D.W. Klemm (rklemm@jhsph.edu)
Mahbubur Rashid (mrashid.jivita@gmail.com)
Abu Ahmed Shamim (shamim.jivita@gmail.com)
Allan Massie (amassie@jhsph.edu)
Kerry Schulze (kschulze@jhsph.edu)
Andre Hackman (ahackman@jhsph.edu)
Keith P West Jr (kwest@jhsph.edu)

Version: 3 Date: 1 April 2011

Author's response to reviews: see over
Dear Editors:

We are pleased to enclose the revised manuscript entitled “A Cluster-Randomized, Placebo-controlled, Maternal Vitamin A or Beta-Carotene Supplementation Trial in Bangladesh: Design and Methods”, assigned MS# 1959845324473273. We thank the editors and reviewers for their careful feedback, which we have incorporated into this revision. The following specific concerns were addressed as follows:

Editor:
1. Authors' Contributions, Competing Interests, and Acknowledgments sections, have been moved to the end of the manuscript after the Conclusions and before the reference list.
2. The Abstract has been restructured to contain four sections: Background, Method, Results and Conclusion and is now no longer than 350 words.
3. The main manuscript text is now divided into the following sections: Background, Method, Results, Discussion and Conclusion.
4. A list of abbreviations is now included at the end of the manuscript before the reference list, including all abbreviations used in the manuscript and their meanings.

All contributing authors have reviewed and approved the content of this revised manuscript. We thank you in advance for your positive consideration of this manuscript and will be glad to respond to any further queries or suggestions you may have regarding this submission.

Sincerely,

Alain B. Labrique, MS, MHS, PhD
Assistant Professor